Author's response to reviews

Title: Costs and risk factors for ventilator-associated pneumonia in a Turkish University Hospital's Intensive Care Unit: A case-control study

Authors:

Dr Hakan R Erbay (rherbay@pamukkale.edu.tr)
Prof Nevzat A Yalcin (anyalcin@yahoo.com)
Mehmet Zencir (mzencir@pamukkale.edu.tr)
Prof Simay Serin (sserin@pamukkale.edu.tr)
Habip Atalay (hatalay@pamukkale.edu.tr)

Version: 4 Date: 22 Apr 2004

Dear Editors;

We revised our manuscript according to your format directions as follows:

1. Title page: The email addresses for all authors were included in this page.

2. Abstract: It has been structured into Background, Methods, Results and Conclusions.

3. Keywords section: The "ICU" abbreviation was extracted, and "intensive care unit" replaced in this section.

4. Introduction: This section was renamed as Background.

5. References: References were cited in the text using consecutive numbers in square brackets. Citations in the reference list contains all names of the authors.

6. Tables- The title list was removed from the text.

Sincerely

Dr.R.Hakan Erbay